Author’s response to reviews

Title: Protocol: Randomised clinical trial investigating the effectiveness and cost benefit of a lifestyle intervention targeting Type 2 Diabetes in Australia.

Authors:
Linda Cloete (linda.cloete@avondale.edu.au)
Brett Mitchell (brett.mitchell@avondale.edu.au)
Darren Morton (darren.morton@avondale.edu.au)

Version: 4 Date: 07 Jun 2019

Author’s response to reviews:

Thank you for considering to publish our protocol paper “Randomised clinical trial investigating the effectiveness and cost benefit of a lifestyle intervention targeting Type 2 Diabetes in Australia.”

As requested, below is the response to the requests made.

1. Objectives removed as requested

2. Role of funding bodies added

3. Highlighting removed of previous edits and document passed checking proves for annotations etc.

4. This was already present in the text at the end of page 10